Author's response to reviews

Title: Acute dengue virus 2 infection in Gabonese patients is associated with an early innate immune response, including strong interferon alpha production

Authors:

Pierre Becquart (pierre.becquart@ird.fr)
Nadia Wauquier (wauquiernadia@orange.fr)
Dieudonné Nkoghe (dnkoghe@hotmail.com)
Angélique Ndjoyi-Mbiguino (ainanga@yahoo.fr)
Cindy Padilla (padilla.cindy@gmail.com)
Marc Souris (marc.souris@ird.fr)
Eric Leroy (eric.leroy@ird.fr)

Version: 3 Date: 13 September 2010

Author's response to reviews: see over